Author’s response to reviews

Title: Validation of a Self-Administered Questionnaire to Assess Bullying in a Highly Violent Mexican Communities

Authors:

Arnulfo Ramos-Jimenez (aramos@uacj.mx)
Abraham Wall-Medrano (awall@uacj.mx)
Oscar Esparza-Del Villar (reducible01@uacj.mx)
Rosa P Hernandez-Torres (rphernant@yahoo.com)

Version: 3 Date: 6 March 2013

Author’s response to reviews:

March 5th, 2013

Prof. Inga Dora Sigfusdottir
Associate Editor
BMC Public Health

We send to your consideration the final revised manuscript entitled: Design and Validation of a Self-Administered Test to Assess Bullying (Bull-M) in High School Mexicans: A Pilot Study, originally entitled and registered as Validation of a self-administered questionnaire to assess bullying in a highly violent community (MS: 186519061814151). We have addressed the following comments from Reviewer 2:

Reviewer 2 (Alfgeir Logi Kristjansson)

A) Main concerns

1. (the manuscript) needs some language corrections before being published.

R. A third native English speakers edited the manuscript, appendix, tables and figures.

2. Change the repetitive wording of “Linkert” as “Likert”

R. Modified accordingly

Sincerely

Arnulfo Ramos-Jiménez, PhD
Universidad Autónoma de Ciudad Juárez – Instituto de Ciencias Biomédicas.
Anillo Envolvente del Pronaf y Estocolmo s/n. C.P. 32300.
Tels. +52 (656) 688-1800, Fax (656) 688-1821.
Ciudad Juárez (32300) Chihuahua, México.